%% LyX 1.3 created this file.  For more info, see http://www.lyx.org/.
%% Do not edit unless you really know what you are doing.
\documentclass[english,conference]{IEEEtran}
\usepackage[T1]{fontenc}
\usepackage[latin1]{inputenc}
\usepackage{float}
\usepackage{amsmath}
\usepackage{graphicx}
\usepackage{setspace}
\usepackage{amssymb}

\makeatletter

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% LyX specific LaTeX commands.
\floatstyle{ruled}
\newfloat{algorithm}{tbp}{loa}
\floatname{algorithm}{Algorithm}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% User specified LaTeX commands.

\usepackage{cite}

\usepackage{babel}
\makeatother
\begin{document}

\title{{\Large Medical Image Processing Related Work}\\
{\Large Skin Lesion}}


\author{Howard Zhou}

\maketitle

\section{\label{sec:Related-work}Related work}

\begin{enumerate}
\item Skin Lesion: \cite{Cucchiara:2002:Iterative},
\cite{Fischer:1996:Analysis}, \cite{Schmid:1997:Colour},
\cite{Ganster:2001:Automated}, \cite{Schmid:1999:Thesis},
\cite{Schmid:2003:Towards}, \cite{Fleming:1998:Techniques},
\cite{Ercal:1993:Detection}, \cite{Chung:2001:Content-based},
\cite{Celebi:2004:Content-Based}, \cite{Anantha:2004:Detection},
\cite{Roberts:2003:Artificially}, \cite{Umbaugh:1991:Applying},
\cite{Huyser:2004:Discovering}, \cite{Carrara:2005:Automated},
\cite{Dreiseitl:2001:Comparison}, \cite{Sun:1999:Fast},
\cite{Jailani:2004:Border}, 

\end{enumerate}



\bibliographystyle{plain}
\bibliography{related_work}

\end{document}
